

The potential benefit of statin prescription based on prediction of treatment responsiveness in older individuals: an application to the PROSPER randomized controlled trial

Nguyen, T.L.; Trompet, S.; Brodersen, J.B.; Hoogland, J.; Debray, T.P.A.; Sattar, N.; ... ; Westendorp, R.G.J.

Citation

Nguyen, T. L., Trompet, S., Brodersen, J. B., Hoogland, J., Debray, T. P. A., Sattar, N., ... Westendorp, R. G. J. (2023). The potential benefit of statin prescription based on prediction of treatment responsiveness in older individuals: an application to the PROSPER randomized controlled trial. *European Journal Of Preventive Cardiology*, *31*(8), 945-953. doi:10.1093/eurjpc/zwad383

Version:Publisher's VersionLicense:Creative Commons CC BY 4.0 licenseDownloaded from:https://hdl.handle.net/1887/3764031

Note: To cite this publication please use the final published version (if applicable).



The potential benefit of statin prescription based on prediction of treatment responsiveness in older individuals: an application to the PROSPER randomized controlled trial

Tri-Long Nguyen (1)¹*, Stella Trompet^{2,3}, John B. Brodersen (1)^{4,5}, Jeroen Hoogland (1)^{6,7}, Thomas P.A. Debray (1)^{6,8}, Naveed Sattar (1)⁹, J. Wouter Jukema (1)^{2,10}, and Rudi G.J. Westendorp^{1,11}

¹Section of Epidemiology, Department of Public Health, University of Copenhagen, Øster Farimagsgade 5, DK-1356 Copenhagen K, Denmark; ²Department of Cardiology, Leiden University Medical Centre, Leiden, The Netherlands; ³Departments of Gerontology and Geriatrics, Leiden University Medical Centre, Leiden, The Netherlands; ⁴Centre of General Practice, Department of Public Health, University of Copenhagen, Copenhagen, Denmark; ⁵Primary Health Care Research Unit, Region Zealand, Denmark; ⁶Julius Center for Health Sciences and Primary Care, University Medical Center Utrecht, Utrecht University, Utrecht, The Netherlands; ⁷Department of Epidemiology and Data Science, Amsterdam University Medical Centers, Amsterdam, The Netherlands; ⁸Smart Data Analysis and Statistics B.V., Utrecht, The Netherlands; ⁹School of Cardiovascular & Metabolic Health, British Heart Foundation Centre of Research Excellence for Heart Failure Prevention and Treatment, University of Glasgow, Glasgow, United Kingdom; ¹⁰Durrer Center for Cardiovascular Research, Netherlands Heart Institute, Utrecht, The Netherlands; and ¹¹Center for Health Ageing, University of Copenhagen, Copenhagen, Denmark

Received 8 November 2022; revised 3 March 2023; accepted 6 December 2023; online publish-ahead-of-print 12 December 2023

Aims	Clinical guidelines often recommend treating individuals based on their cardiovascular risk. We revisit this paradigm and quantify the efficacy of three treatment strategies: (i) <i>overall</i> prescription, i.e. treatment to all individuals sharing the eligibility criteria of a trial; (ii) <i>risk-stratified</i> prescription, i.e. treatment only to those at an elevated outcome risk; and (iii) prescription based on predicted <i>treatment responsiveness</i> .
Methods and results	We reanalysed the PROSPER randomized controlled trial, which included individuals aged 70–82 years with a history of, or risk factors for, vascular diseases. We conducted the derivation and internal–external validation of a model predicting treatment responsiveness. We compared with placebo ($n = 2913$): (i) pravastatin ($n = 2891$); (ii) pravastatin in the presence of previous vascular diseases and placebo in the absence thereof ($n = 2925$); and (iii) pravastatin in the presence of a favourable prediction of treatment response and placebo in the absence thereof ($n = 2890$). We found an absolute difference in primary outcome events composed of coronary death, non-fatal myocardial infarction, and fatal or non-fatal stroke, per 10 000 person-years equal to: -78 events (95% Cl, -144 to -12) when prescribing pravastatin to all participants; -66 events (95% Cl, -114 to -18) when treating only individuals with an elevated vascular risk; and -103 events (95% Cl, -162 to -44) when restricting pravastatin to individuals with a favourable prediction of treatment response.
Conclusion	Pravastatin prescription based on predicted responsiveness may have an encouraging potential for cardiovascular preven- tion. Further external validation of our results and clinical experiments are needed.
Trial registration	ISRCTN40976937.
Lay summary	 This study investigates whether an algorithm to predict how much old age individuals would benefit from a statin treatment could be useful to guide clinicians in their prescription decision-making; the key findings are as follows: About one out of seven individuals included in the study has no predicted benefit of pravastatin. Compared with prescribing pravastatin to all old age individuals at risk of cardiovascular diseases, withholding pravastatin in those with no predicted benefit seems to lead to a better prevention of cardiovascular events.

^{*} Corresponding author. Tel: +45 35335402, Email: long@sund.ku.dk

[©] The Author(s) 2023. Published by Oxford University Press on behalf of the European Society of Cardiology. All rights reserved. For permissions, please e-mail: journals.permissions@oup.com

Graphical Abstract



Introduction

In the early 1990s, the concept of 'evidence-based medicine' (EBM) set a new foundation for treatment decision-making based on results preferably drawn on systematic studies of randomized controlled trials (RCTs).¹ Refinements of this paradigm rapidly emerged over the years, as some started questioning the applicability of the overall results of RCTs to patient treatment in clinical practice: while the RCT provides the best available evidence of the *average* effects of treatments, patients with different characteristics and risk profiles might display heterogeneous responses to treatments.^{2–4} For instance, lowering cholesterol with statins decreases the risk of cardiovascular outcomes *on average*; yet, a person at low risk of cardiovascular disease may not benefit from this preventive treatment as much as a patient with a known history of cardiovascular disease and multiple chronic comorbidities.

In response to this issue, major contributions of the last decade have sharpened the initial concept of overall EBM to risk-stratified EBM⁵⁻⁹; the results of RCTs becoming tailored to the patient risk profile and showing, for instance, that patients at a high risk of poor outcome are likely to have a more beneficial treatment response than patients at a low risk.¹⁰ However, a limitation of such a risk stratification may reside in the patient risk being explained in fact by different compositions of factors; for instance, a same high level of cardiovascular risk can be due to uncontrolled hypertension for one patient, and to advanced-stage diabetes for another. If those factors modify the treatment effect in different magnitudes and directions, e.g. suppose a treatment is highly effective in people with hypertension but ineffective in people with diabetes, the advantage of applying risk-stratified EBM to treatment decisionmaking becomes uncertain: on the one hand, treating only patients at high risk may lead to prescribing treatments to some for whom it is ineffective (i.e. overtreatment); on the other hand, leaving patients at low

risk untreated may lead to neglecting an effective treatment for those who could benefit from it (i.e. undertreatment).

Instead of making a treatment decision based on the risk level, one may want to consider anticipated clinical responsiveness to treatment (i.e. benefit-to-harm balance) as the main criterion. The strength of such an approach may be two-fold, both by encouraging treatment prescription when benefits are expected and by suggesting no prescription, or deprescription, when harms or no benefits are predicted. For instance, while the benefit of widely prescribed statins might be expected in the majority of individuals, it is also crucial to anticipate treatment intolerance,¹¹ or side effects, such as the increase of diabetes risk.¹² In the population of older individuals, statin prescription strategy for cardiovascular prevention may not be straightforward, given the limitations of cardiovascular risk scoring systems [e.g. Systematic Coronary Risk Estimation (SCORE) applicable only to individuals aged \leq 70 years], and little evidence of statin benefit in individuals aged > 75 years without history of vascular disease (see ESC/EAS guidelines 2019 for the management of dyslipidaemias).¹³ Recent examples in cardiovascular diseases showed how the development and validation of clinical prediction models could help clinicians identify patients who may, or may not, benefit from pharmacological treatment, by predicting the responsiveness to drug therapies. $^{14-16}$ In this article, we intend to further demonstrate the importance of moving the paradigm on risk stratification towards predicted treatment responsiveness. As an application, we reanalyse the data of the PROSPER study: a major randomized controlled trial showing evidence of the overall protective effect of pravastatin treatment amongst older individuals at risk of vascular disease.¹⁷ For the clinician, the issue of the prediction of individualized treatment responsiveness can be formulated as the shift in patient prognosis due to pravastatin prescription (i.e. the contrast in cardiovascular risk if the patient were to be prescribed pravastatin vs. if the patient were not to be prescribed

pravastatin). Such a prediction can be obtained by modelling the prognosis of individuals sharing similar characteristics (in terms of sex, lifestyle, and medical history) and who have been exposed to either pravastatin or placebo, respectively.

In this reanalysis, we aim to quantify the effect sizes of three different treatment strategies: (i) an overall strategy, i.e. prescribing pravastatin to all old individuals sharing the characteristics of those recruited in the experiment; (ii) a *risk-stratified* strategy, i.e. prescribing pravastatin only to those who have suffered a previous vascular event (secondary or tertiary prevention); and (iii) an individualized *treatment responsiveness* strategy, i.e. prescribing pravastatin only to those for whom a favourable outcome is predicted under statin treatment.

Methods

Data and participants

The PROSPER trial was a multicentre, randomized placebo-controlled trial evaluating the effect of pravastatin on major vascular events in older individuals with pre-existing vascular disease or increased risk (e.g. smokers, hypertension, or diabetes people).¹⁷ The trial included 5804 individuals aged 70–82 years from Scotland, Ireland, and The Netherlands between 1997 and 1999.¹⁷ Additional information about the PROSPER trial can be found elsewhere (e.g. see original report and protocol).^{17,18}

The PROSPER trial complied with the Declaration of Helsinki; written informed consents were obtained.¹⁷ The ethics committees of all centres approved the original study. Data sharing agreements can be made upon reasonable request to the investigators of the trial.

Outcomes

The primary outcome was defined as a composite outcome of coronary death, non-fatal myocardial infarction, and fatal or non-fatal stroke.¹⁷ In this reanalysis, we included the following as secondary outcomes: all-cause mortality, non-cardiovascular death, cardiovascular death, fatal stroke, and coronary heart disease death. A detailed definition of these outcomes can be found in the original protocol of the trial.¹⁸

Statistical analysis

To guide the treatment prescription, we derived a model predicting the responsiveness to pravastatin and assessed its performance. For a given person, the endpoint to be predicted was not the occurrence of the primary outcome *per* se, but the *responsiveness* to pravastatin—also referred to as 'individualized treatment effect'^{19,20}—that is, the contrast in potential primary outcome under two scenarios: the probability of event if they were to receive pravastatin vs. the probability of event if they were to receive placebo.

We took advantage of the multicentre nature of the PROSPER trial to perform 'internal-external cross-validation' (IECV), which is a recent, attractive method for both deriving generalizable prediction models and assessing how well they would likely perform in practice, using large, clustered data.²¹⁻²⁵ Briefly, when performing IECV in data including K centres, one out of the K centres is left out to assess the validity of a prediction model derived on the remaining K-1 centres. This procedure is repeated by rotating the holdout centres, thus allowing one to assess the generalizability of a prediction model K times, across K centres, where practice, measurement, and case-mix are likely to differ and affect the predictive performance. Throughout IECV, we fit a prediction model using Cox proportional hazard regression, including as explanatory variables the treatment status (i.e. pravastatin/placebo) and the following baseline covariates: age, sex, smoking status, baseline cholesterol and triglyceride levels, prior comorbidities, treatments (others than statins), and cognitive function [mini-mental state examination (MMSE)]. Based on clinical knowledge and the numerous existing prediction models of cardiovascular risk,²⁶ these variables were assumed to be likely predictors of the primary outcome. We included two-way interaction terms between the treatment status and these baseline covariates, which were also considered as highly credible potential effect modifiers. We applied penalization by least absolute shrinkage and selection operator (lasso) to reduce the risk of over-fitting predicted individualized treatment response.^{20,27-30} We used IECV to identify the combination of interaction terms leading to the

prediction model with the best performance at validation in the hold-out clusters (see Supplementary material online, Appendix S1A and S1B).

We defined the performance as the potential clinical impact of the prediction model, that is, the reduction in primary outcome associated with the treatment rule guided by the predicted treatment responsiveness. The prediction model using Cox regression returned for each person a predicted hazard under pravastatin and a predicted hazard under placebo, the ratio of the two corresponding to an individualized hazard ratio (iHR). We constructed a treatment rule based on this predicted responsiveness,³¹ which defined individuals to be prescribed pravastatin as those with iHR < 1 (so-called 'treatment-favourable' individuals likely to benefit from pravastatin), and individuals to be prescribed no pravastatin (placebo) as those with $iHR \ge 1$ ('treatment-unfavourable' individuals unlikely to benefit from pravastatin). We computed the reduction in primary outcome associated with this treatment rule (see Supplementary material online, Appendix S1B). The greater this reduction, the greater the clinical benefit of the treatment decision strategy based on the prediction model. Thus, we identified throughout IECV the 'best' candidate prediction model with interaction terms that led to optimizing the clinical benefit of treatment decision at validation.

In a final step, we fit the 'best' candidate model to the full sample to make use of all data available at hand, applying a lasso penalization to the estimation of the model coefficients. We used this final model to identify 'treatmentfavourable' and 'treatment-unfavourable' individuals in the full sample. We assessed the effect of the optimal treatment rule based on these predicted 'treatment-favourable'/'treatment-unfavourable' profiles on the primary outcome.³¹ We estimated in the full sample the effect sizes of three decision strategies allocating treatment to (1) all, (2) those at high risk, and (3) those with predicted favourable treatment responsiveness. Group 1 included all individuals randomly allocated to pravastatin treatment (i.e. initial treatment arm); Group 2 included individuals with history of vascular disease randomly allocated to pravastatin treatment and individuals without history of vascular disease randomly allocated to placebo; and Group 3 included 'treatmentfavourable' individuals randomly allocated to pravastatin treatment and 'treatment-unfavourable' individuals randomly allocated to placebo. An estimation of the effect sizes associated to each of the three groups was performed as an intention-to-treat analysis following randomization (see Supplementary material online, Appendix S1C). We defined a common reference as the group of all individuals not allocated to pravastatin (i.e. initial placebo arm). We estimated the effect sizes on an absolute scale [hazard difference (HD)] using Aalen additive hazard model, and on a relative scale [hazard ratio (HR)] using Cox proportional hazard model with Firth penalization.³² We computed confidence intervals by bootstrapping (1000 iterations).

Overall, our analysis followed the Predictive Approach to Treatment Heterogeneity (PATH) statement^{7,9} in limiting the set of variables to be included in the prediction model to highly credible effect modifiers, performing no one-variable-at-a-time selection, adopting penalized regression, and defining a clinically meaningful performance measure instead of conventional metrics of discrimination and calibration of predicted risk.

No imputation method was performed, given the very low rate of missing data regarding the aforementioned variables (1.2%). All statistical analyses were performed in R, version 3.6.2 (further details on statistical packages, estimation, and regression techniques are in Supplementary material online, Appendix S1).

Role of the funding source

The funding sources had no roles in the study design and in the collection, analysis, and interpretation of data nor in the writing of the manuscript and in the decision to submit it for publication.

Results

The primary outcome occurred in 473/2913 (16.2%) and 408/2891 (14.1%) individuals randomly allocated to treatment with placebo and pravastatin, respectively: HR = 0.85 (95% Cl 0.74–0.97), which was the effect size of the *overall* treatment strategy reported in the original analysis of the PROSPER trial. On an absolute scale, this corresponded to a hazard difference of -78 events (95% Cl, -144 to -12) per 10 000 person-years.

When applying a *risk-stratified* strategy initiating pravastatin treatment only in old individuals with a history of vascular disease and leaving Table 1Coefficients and baseline cumulative hazardfunctions of the model predicting the rate of primaryoutcome under two hypothetical scenarios: ifpravastatin were prescribed and if pravastatin were notprescribed

Baseline characteristics	Coefficient		
	lf pravastatin prescribed	lf pravastatin not prescribed	
Provoctatio		0.0011	
A ray (vegera)		-0.0011	
Age (years)	0.0392	0.0372	
Smolver	0.3011	0.3011 - 0.1041 0.1754 \pm 0.1175	
	0.1754	0.175++0.1175	
	0.0177	0.0177	
HDL-cholesterol	-0.0177	-0.0177	
Care arbidities	0.0066	0.0066	
Comorbidities:	0.0641	0.0641	
History of hypertension	0.0641	0.0641	
History of diabetes	0.3779	0.3779	
History of vascular	0.0774	0.0774-0.0360	
disease	0 4272	0 4272	
History of myocardial	0.4373	0.4373	
	0.0257	0.0257	
	-0.0237	-0.0237	
History of PTCA	0.0000	0.0000	
History of algudiantian	0.2122	0.2122	
History of Claudication	0.4706	0.4706	
History of PVD surgery	-0.3184	-0.3184	
ACE includes	0.0275	0.0275 + 0.0022	
ACE Inhibitors	0.0365	0.0365 + 0.0022	
Beta-blockers	0.0981	0.0981	
	0.2017	0.2017	
Other anti-hypertensives	0.5368	0.5368-0.3773	
Aspirin	0.1489	0.1489	
Anti-coagulants	0.1840	0.1840	
Anti-arrhythmic	0.3936	0.3936	
Non-insulin diabetes	-0.1994	-0.1994 + 0.5246	
	-0.0546	-0.0546	
Baseline cumulative	$Ln[H_0(time)] = -5$.1/46 + 1.10/6 Ln(time)	
hazard function			

For each individual, the individualized hazard ratio (iHR) is predicted as follows: iHR = exp(-0.0811 + -0.1641*male sex + 0.1175*smoker + -0.0360*history of vascular disease + 0.0022*ACE inhibitors + -0.3773*other antihypertensive drugs + 0.5246*non-insulin diabetes)

Under each scenario, the hazard is predicted by calculating the product of the baseline cumulative hazard function H0(time) (in years) and the exponential of the linear predictor [i.e. linear predictor = sum (coefficients*predictor values)].

those without history of vascular disease untreated, we observed an absolute difference of -66 events (95% Cl, -114 to -18) per 10 000 person-years. This corresponded to a 12% relative reduction of the primary outcome: HR = 0.88 (95% Cl: 0.80 to 0.97).

We assessed the performance of a prescription strategy based on the prediction of an individualized treatment responsiveness. We reported the estimated coefficients of the prediction model, which was likely the best at distinguishing individuals who would have the primary outcome decreased under pravastatin (i.e. 'treatment-favourable') from those who would not (i.e. 'treatment-unfavourable') based on six treatment effect modifiers (see Table 1). We found that following the individualized treatment responsiveness strategy, prescribing pravastatin only to those who were predicted to be 'treatment-favourable' and leaving untreated those predicted to be 'treatment-unfavourable', yielded an absolute difference of -103 events (95% CI: -162 to -44) per 10 000 person-years, that is, a 19% relative reduction [HR = 0.81 (95% Cl: 0.71 to 0.91)], which seemed to outperform the overall and risk-stratified treatment strategies (Figure 1). In a formal comparison, the individualized treatment responsiveness strategy led to an absolute difference of -23 events (95%) CI: -47 to 1) per 10 000 person-years compared with the overall strategy [5% relative reduction of the primary outcome, HR = 0.95 (95% CI: (0.90 to 1.00) and to an absolute difference of -37 events (95% CI: -76to 2) per 10 000 person-years compared with the risk-stratified strategy [8% relative reduction, HR = 0.92 (95% CI: 0.85 to 1.00)]. To prevent possible optimism in the estimated effect sizes, we performed a bootstrap-correction procedure taking into account the uncertainty of all sequences of the analysis (see Supplementary material online, Appendix S1D). The results were similar, indicating no optimism (see Supplementary material online, Appendix S2B).

As depicted in *Figure 2*, the prediction model predicted pravastatin to provide benefit in a majority of old age people (86.1%); the rest (13.9%) predicted to have no benefit. We reported in *Table 2* the baseline characteristics of the 'treatment-favourable' and 'treatment-unfavourable' individuals. The group of 'treatment-favourable' individuals included on average more male, non-smoking individuals with history of hypertension, vascular disease, myocardial infarction, and angina. In addition, they were likely to be non-diabetic.

Conventional subgroup analyses based on treatment responsiveness indicated a clear protective effect of pravastatin on the primary outcome within individuals who were predicted to be 'treatment-favourable' [HD = -122 events (95% CI: -193 to -51) per 10 000 person-years; HR = 0.78 (95% CI: 0.68 to 0.90)]. Conversely, within individuals predicted to be 'treatment-unfavourable', there was an increase of the primary outcome under pravastatin compared with placebo [HD = 187 events (95% CI: 2 to 372) per 10 000 person-years; HR = 1.43 (95% CI: 1.01 to 2.05)].We explicitly emphasize caution in interpreting this subgroup analysismade on 13.9% of the sample, which may be the result of chance finding.The results are reported for the sake of transparency only. The subgroupanalyses for the secondary outcomes are summarized in*Table 3*.

We explored whether the 'treatment-favourable' and 'treatmentunfavourable' profiles were merely associated with adherence and non-adherence behaviours rather than the clinical responsiveness to pravastatin; we found no significant association (see Supplementary material online, Appendix S2C).

We also investigated whether the 'treatment-favourable' and 'treatment-unfavourable' profiles were associated with the change to baseline LDL-cholesterol (at 3, 6, 12, 24, and 36 months) under treatment with pravastatin; we found a very minor further reduction of LDL-cholesterol levels in 'treatment-favourable' individuals compared with 'treatment-unfavourable' individuals (see Supplementary material online, Appendix S2C).

Discussion

Using a newly derived and validated prediction model for estimating treatment responsiveness, we have identified a large subgroup of old age individuals who are likely to benefit from pravastatin, and on the other hand, a minority of old age individuals likely to have no benefit from this treatment. Our main results indicate that prescribing pravastatin only to old people who were predicted to have a favourable response under pravastatin may be a more effective strategy for reducing cardiovascular outcomes than prescribing pravastatin only to those at high risk (as those who suffered a previous clinical event), or to prescribing pravastatin to all older people.



Figure 1 Effect sizes of three evidence-based decision strategies for prescribing pravastatin compared with placebo treatment: (i) prescription to all; (ii) prescription only for those at risk defined by a history of vascular disease; and (iii) prescription only to those expected to have a favourable treatment response defined by a predicted individualized hazard ratio (iHR) < 1. Hazard differences and hazard ratios (with 95% confidence intervals) are reported for the primary outcome: coronary death, non-fatal myocardial infarction, and fatal or non-fatal stroke.





Table 2Characteristics across individuals predicted to benefit from pravastatin (i.e. 'treatment-favourable' individuals
with iHR < 1) and individuals predicted to have no benefit from pravastatin (i.e. 'treatment-unfavourable' individuals
with iHR \geq 1). Means (standard deviations) and counts (percentages) are reported for continuous and categorical
variables, respectively. P-values are not reported, as the non-random model-based subgrouping inherently eliminates
the null hypothesis

Baseline characteristics	'Treatment-favourable' n = 4935 (86.1%)	'Treatment-unfavourable' n = 800 (13.9%)
Age (years)	75.40 (3.37)	74.91 (3.17)
Male sex	2581 (52.3)	189 (23.6)
Smoker	1042 (21.1)	495 (61.9)
Lipid levels:		
LDL-cholesterol	3.81 (0.79)	3.73 (0.83)
HDL-cholesterol	1.27 (0.34)	1.31 (0.39)
Total cholesterol	5.69 (0.91)	5.67 (0.90)
Triglycerides	1.54 (0.71)	1.59 (0.67)
Comorbidities:		
History of hypertension	3225 (65.3)	329 (41.1)
History of diabetes	262 (5.3)	355 (44.4)
History of vascular disease	2402 (48.7)	128 (16.0)
History of myocardial infarction	715 (14.5)	50 (6.2)
History of CABG	142 (2.9)	9 (1.1)
History of PTCA	92 (1.9)	7 (0.9)
History of angina	1465 (29.7)	74 (9.2)
History of claudication	355 (7.2)	25 (3.1)
History of PVD surgery	109 (2.2)	11 (1.4)
Treatments:		
ACE inhibitors	821 (16.6)	117 (14.6)
Angiotensin II receptor blockers	96 (1.9)	17 (2.1)
Beta-blockers	1374 (27.8)	111 (13.9)
Calcium channel blockers	1321 (26.8)	124 (15.5)
Other anti-hypertensives	222 (4.5)	11 (1.4)
Nitrates	1018 (20.6)	61 (7.6)
Aspirin	1914 (38.8)	160 (20.0)
Anti-coagulants	103 (2.1)	10 (1.2)
Anti-arrhythmics	135 (2.7)	19 (2.4)
Non-insulin diabetes	8 (0.2)	348 (43.5)
Insulin diabetes	43 (0.9)	8 (1.0)
MMSE	28.04 (1.53)	27.93 (1.63)

The overall results of RCTs, including the PROSPER trial,¹⁷ have shown that treating old people with statins irrespective of their risk of vascular disease provides benefit-both from utilitarian and deontological perspectives. Although statins may provide protection in the majority of old individuals, our reanalysis of the PROSPER trial shows that pravastatin prescription lacks a favourable outcome in about one out of seven individuals. While it may be acceptable to overtreat a small portion of individuals of the general population given the low-risk safety profile and important benefits of statins, concerns are different in old age. Polypharmacy in elderly individuals is known to be a major health issue that deserves attention and action; we believe it would be unethical to continue a treatment that could be safely withheld in some individuals. To help clinicians identify old age individuals who are likely to benefit from pravastatin, we report a model predicting the responsiveness to pravastatin on vascular events. Estimating the effect sizes of different treatment decision strategies based on risk assessment or prediction models is pivotal to informing clinicians; yet, it is never or rarely done.

Thus, we do not recommend interpreting the subgroup analysis reported in Figure 2 alone; rather, our main analysis aims to infer the overall effect of prescription strategies (Figure 1). In this study, we have assessed the potential clinical efficacy of the prescription strategy guided by our prediction model, using novel methods.³¹ Unlike conventional subgroup results—which are still recommended for they provide a rough guidance to clinicians^{7,9}—we quantified the overall benefit-to-harm balance of treating some individuals while leaving others untreated following different prescription decision rules. This allowed us to infer the overall effect of decision rules, even though they may misclassify individuals who should be treated or not treated. For instance, it is likely that our prediction was not perfect, and that some individuals labelled 'favourable' were in fact 'unfavourable' and vice versa; the impact of such possible misclassification was accounted for in the estimate of the overall effect of the prescription strategy. Although the benefit of initiating treatment with pravastatin in older people without a previous history of vascular disease is still under debate,¹³ our results indicate that disregarding such a Table 3Subgroup analyses of the primary and secondary outcomes within individuals predicted to benefit from
pravastatin (i.e. 'treatment-favourable' individuals with iHR < 1) and individuals predicted to have no benefit from
pravastatin (i.e. 'treatment-unfavourable' individuals with iHR \geq 1). Hazard differences (with 95% confidence intervals)
and hazard ratios (with 95% confidence intervals corrected by Firth penalization) are reported, along with P-values for
additive and multiplicative interaction tests, respectively

Outcome	'Treatment-favourable' Hazard difference ^a	'Treatment-unfavourable' Hazard difference ^a	Interaction test P-value Additive scale
Primary outcome	-122.0 (-193.0 to -51.0)	187.0 (2.2 to 372.0)	0.002
All-cause mortality	-38.6 (-93.3 to 16.1)	226.0 (65.1 to 387.0)	0.002
Non-cardiovascular death	3.3 (-35.4 to 41.9)	135.0 (9.6 to 260.0)	0.052
Cardiovascular death	-41.9 (-80.5 to -3.3)	90.9 (-10.0 to 192.0)	0.014
Fatal stroke	3.6 (-9.2 to 16.4)	44.1 (-4.1 to 92.3)	0.109
CHD death	-38.9 (-72.2 to -5.6)	21.6 (-62.5 to 106.0)	0.161
	Hazard ratio	Hazard ratio	Multiplicative scale
Primary outcome	0.78 (0.68 to 0.90)	1.43 (1.01 to 2.05)	0.002
All-cause mortality	0.88 (0.74 to 1.05)	1.73 (1.18 to 2.58)	0.002
Non-cardiovascular death	1.02 (0.80 to 1.31)	1.72 (1.05 to 2.86)	0.066
Cardiovascular death	0.76 (0.59 to 0.98)	1.74 (0.95 to 3.30)	0.011
Fatal stroke	1.23 (0.58 to 2.63)	3.40 (0.91 to 18.17)	0.198
CHD death	0.71 (0.53 to 0.95)	1.20 (0.59 to 2.49)	0.155

^aHazard differences are expressed in events per 10 000 person-years.

preventive treatment may result in a loss of overall benefit of pravastatin therapy. Following the suggestion of a reviewer, we performed an additional analysis based on the recent SCORE2-OP developed and validated by de Vries et al. (2021).³³ Although the SCORE2-OP is valid to predict the 10-year risk of cardiovascular events in elderly individuals, we found that it did not perform well in indicating who would benefit from pravastatin and who would not benefit from this treatment (see Supplementary material online, Appendix S2D). These additional results are not at odds with de Vries et al.³³ who found no interaction between the effect of pravastatin and the SCORE2-OP (see their Supplementary Material).

While some have warned against the confusion between nonresponse and non-adherence to treatment,³⁴ our reanalysis shows that non-adherence is not associated with treatment responsiveness as predicted by our model. Neither the adherence to nor the reduction in LDL-cholesterol under pravastatin treatment is strongly associated with the predicted responsiveness. Both explain only little of the benefit amongst those who are predicted to have a favourable response to pravastatin. These results are not at odds with the findings that monitoring vascular risk in old age based on cholesterol may not suffice.^{35,36} In a recent meta-analysis, the Cholesterol Treatment Trialists' Collaboration (2019) observed with increasing age a trend towards smaller proportional risk reductions in major coronary events from LDL lowering.³⁷ Further, from a prediction perspective, previous literature showed that LDL-cholesterol level was a poor risk indicator of cardiovascular death in old age (e.g. see Weverling-Rijnsburger et al. 2003).³⁸ The diverse, individual protective effect of pravastatin on clinical outcomes may be better predicted by considering other characteristics of the individual profile (e.g. diabetes and hypertension), which our prediction model may help to identify.

The use of prediction models has been advocated to guide clinical decision-making when treatments may display heterogeneous effects across different individual profiles.^{7,9} Ideally, decision-making should be multi-dimensional and take into account multiple endpoints, such as risk for mortality, risk for non-fatal outcomes, risk for side effects, limited resources, and financial constraints. In this reanalysis, we defined

the benefit/harm according to only the cardiovascular events used as primary outcomes in the PROSPER trial; this is a limitation of our study. As the identification of benefit, harm or null effect directly depends on model-based predictions; a key challenge to such an approach is a careful modelling of treatment effect heterogeneity by including relevant interaction terms between predictors and treatment variable.^{27,28} When interactions are not properly handled, the predicted benefit/ harm may be a mere result of over-fitting: in such a case, a seemingly neat distinction between individuals with and without a favourable response to treatment may in fact be over-optimistic.^{27,28} To prevent spurious finding, we adopted a stringent penalization of our prediction model, assessed its generalizability multiple times using internal-external cross-validation, corrected our validation results for potential small sample bias, and performed a final bootstrap-correction procedure of the whole analysis. Limiting the set of effect modifiers to prevent overoptimistic results may have led to omitting possible, not yet identified, important variables and/or interactions.

The clinical prediction model we present could optimize the benefit and minimize the harm of pravastatin treatment by helping clinicians to 'choose wisely' and distinguish those individuals expected to benefit from the treatment from those expected to have no benefit or suffer harm. A key limitation is that the usefulness of our prediction model in daily clinical practice is still to be proven. External validation using other data than the PROSPER study needs to be conducted to establish the generalizability of our model. Further studies are needed, such as a randomized controlled trial comparing strategies of prescribing or deprescribing statins with or without the use of our prediction model. Moreover, because pravastatin was the sole treatment randomized in the PROSPER study, inference about the effect of other treatments (targeting other risk factors than LDL-cholesterol) could not be made: thus. our results cannot inform on what treatments could be prescribed to individuals in whom pravastatin should be withheld. Another limitation of our study is our inability to provide explanation on the mechanism behind the more pronounced responsiveness to pravastatin treatment in some individuals compared with others. Our model is meant to be predictive-and not aetiological-therefore, the coefficients reported are

to be used only to predict the treatment responsiveness, since the variables included in the model were found to be indicators (or predictors) of the response to pravastatin; they must *not* be interpreted as causes of the treatment responsiveness. Our prediction rule answers the question 'who are those individuals favoured by pravastatin treatment?' but not 'why do they respond to pravastatin better than other individuals?' Finally, it is worth clarifying that the prediction of *individualized* treatment responsiveness is obtained by modelling the treatment effect in groups of individuals sharing similar measurable characteristics; strictly speaking, this differs from the *individual* treatment responsiveness, which is inaccessible since a single individual (defined by an infinite set of characteristics) is unique and can be exposed only to either pravastatin or placebo, but never both at the same time.

Supplementary material

Supplementary material is available at European Journal of Preventive Cardiology.

Acknowledgements

T.L.N. expresses his gratitude to Dr A. Rieckmann, Prof N.H. Rod, and Prof T. Lange for their initial input at the conceptualization stage of this research. J.H. acknowledges financial support from the Netherlands Organisation for Health Research and Development (grant 91215058). T.P.A.D. is supported by a project that received funding from the European Union's Horizon 2020 research and innovation programme under ReCoDID grant agreement no. 825746. N.S. is supported by British Heart Foundation Centre of Research Excellence grant RE/18/6/34217. R.G.J.W. is supported by a research grant of Novo Nordisk Fonden: 'Harnessing the Power of Big Data to Address the Societal Challenge of Aging [NNF17OC0027812]'.

Author contributions

T.L.N. is the guarantor of this manuscript. T.L.N. is responsible for the conception and design of the work, analysis and interpretation of data for the work, and wrote first draft of the manuscript. S.T. is responsible for the acquisition and interpretation of data for the work, verification of the underlying data, and critical revision of the first draft for important intellectual content. J.B.B. is responsible for the interpretation of data for the work and critical revision of the first draft for important intellectual content. J.H. is responsible for the analysis and interpretation of data for the work and critical revision of the first draft for important intellectual content. T.P.A.D. is responsible for the analysis and interpretation of data for the work and critical revision of the first draft for important intellectual content. J.W.J. is responsible for the acquisition and interpretation of data for the work, verification of the underlying data, and critical revision of the first draft for important intellectual content. R.G.J.W. is responsible for the conception and design of the work, interpretation of data for the work, and first draft of the manuscript.

All authors gave their final approval of the version to be published, and agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Funding

University of Copenhagen; Netherlands Organisation for Health Research and Development; European Union's Horizon 2020; British Heart Foundation; Novo Nordisk Fonden.

Conflict of interest: N.S. has consulted for Afimmune, Amgen, AstraZeneca, Boehringer Ingelheim, Eli Lilly, Hanmi Pharmaceuticals, Merck Sharp & Dohme, Novartis, Novo Nordisk, Pfizer, and Sanofi and

received grant support paid to his University from AstraZeneca, Boehringer Ingelheim, Novartis, and Roche Diagnostics outside the submitted work. All the other authors declare no conflicts of interest.

Data availability

Data sharing agreements can be made upon reasonable request to the investigators of the trial. Requests must be sent to S. Trompet (S.Trompet@lumc.nl), who will provide access to data. Access to data (de-identified participant data, and data dictionary) will require the investigator support, after approval of a proposal, with a signed data access agreement. Access to data will start after agreement signature until the end date mentioned in the agreement form.

References

- 1. Guyatt G. Evidence-Based medicine. JAMA 1992;268:2420.
- Kent DM, Rothwell PM, Ioannidis JP, Altman DG, Hayward RA. Assessing and reporting heterogeneity in treatment effects in clinical trials: a proposal. *Trials* 2010;11:85.
- Rothwell PM. Can overall results of clinical trials be applied to all patients? *Lancet* 1995; 345:1616–1619.
- Rothwell PM. Treating individuals 2. Subgroup analysis in randomised controlled trials: importance, indications, and interpretation. *Lancet* 2005;365:176–186.
- Kent DM, Hayward RA. Limitations of applying summary results of clinical trials to individual patients: the need for risk stratification. JAMA 2007;298:1209–1212.
- Kent DM, Nelson J, Dahabreh JJ, Rothwell PM, Altman DG, Hayward RA, et al. Risk and treatment effect heterogeneity: re-analysis of individual participant data from 32 large clinical trials. Int J Epidemiol 2016;45:2075–2088.
- Kent DM, Paulus JK, van Klaveren D, D'Agostino R, Goodman S, Hayward R, et al. The predictive approaches to treatment effect heterogeneity (PATH) statement. Ann Intern Med 2019;**172**:35. [published Online First: 2019/11/12].
- Kent DM, Steyerberg E, van Klaveren D. Personalized evidence based medicine: predictive approaches to heterogeneous treatment effects. BMJ 2018:363:k4245.
- Kent DM, van Klaveren D, Paulus JK, D'Agostino R, Goodman S, Hayward R, et al. The predictive approaches to treatment effect heterogeneity (PATH) statement: explanation and elaboration. Ann Intern Med 2019;**172**:35–45.
- Glasziou PP, Irwig LM. An evidence based approach to individualising treatment. BMJ 1995;**311**:1356–1359.
- Bytyçi I, Penson PE, Mikhailidis DP, Wong ND, Hernandez AV, Sahebkar A et al. Prevalence of statin intolerance: a meta-analysis. Eur Heart J 2022;43:3213–3223.
- Sattar N, Preiss D, Murray HM, Welsh P, Buckley BM, de Craen AJM, et al. Statins and risk of incident diabetes: a collaborative meta-analysis of randomised statin trials. *Lancet* 2010;**375**:735–742.
- Mach F, Baigent C, Catapano AL, Koskinas KC, Casula M, Badimon L, et al. 2019 ESC/ EAS guidelines for the management of dyslipidaemias: lipid modification to reduce cardiovascular risk. Eur Heart J 2020;41:111–188.
- Basu S, Sussman JB, Rigdon J, Steimle L, Denton BT, Hayward RA, et al. Benefit and harm of intensive blood pressure treatment: derivation and validation of risk models using data from the SPRINT and ACCORD trials. PLoS Med 2017;14:e1002410.
- Yeh RW, Secemsky EA, Kereiakes DJ, Normand S-LT, Gershlick AH, Cohen DJ, et al. Development and validation of a prediction rule for benefit and harm of dual antiplatelet therapy beyond 1 year after percutaneous coronary intervention. JAMA 2016;315: 1735–1749.
- 16. Takahashi K, Serruys PW, Fuster V, Farkouh ME, Spertus JA, Cohen DJ, et al. Redevelopment and validation of the SYNTAX score II to individualise decision making between percutaneous and surgical revascularisation in patients with complex coronary artery disease: secondary analysis of the multicentre randomised controlled SYNTAXES trial with external cohort validation. *Lancet* 2020;**396**:1399–1412.
- Shepherd J, Blauw GJ, Murphy MB, Bollen ELEM, Buckley BM, Cobbe SM, et al. Pravastatin in elderly individuals at risk of vascular disease (PROSPER): a randomised controlled trial. *Lancet* 2002;**360**:1623–1630. [published Online First: 2002/11/30].
- Shepherd J, Blauw GJ, Murphy MB, Cobbe SM, Bollen ELEM, Buckley BM, et al. The design of a prospective study of pravastatin in the elderly at risk (PROSPER). Am J Cardiol 1999;84:1192–1197.
- Nguyen TL, Collins GS, Landais P, Le Manach Y. Counterfactual clinical prediction models could help to infer individualized treatment effects in randomized controlled trials an illustration with the international stroke trial. *J Clin Epidemiol* 2020;**125**:47–56. [published Online First: 2020/05/29].
- Hoogland J, IntHout J, Belias M, Rovers MM, Riley RD, Moons KGM, et al. A tutorial on individualized treatment effect prediction from randomized trials with a binary endpoint. Stat Med 2021;40:5961–5981.

- de Jong VMT, Moons KGM, Eijkemans MJC, Riley RD, Debray TPA. Developing more generalizable prediction models from pooled studies and large clustered data sets. *Stat Med* 2021;40:3533–3559.
- Debray TP, Moons KG, Ahmed I, Koffijberg H, Riley RD. A framework for developing, implementing, and evaluating clinical prediction models in an individual participant data meta-analysis. *Stat Med* 2013;**32**:3158–3180.
- Royston P, Parmar MK, Sylvester R. Construction and validation of a prognostic model across several studies, with an application in superficial bladder cancer. *Stat Med* 2004; 23:907–926.
- Steyerberg EW, Harrell FE Jr. Prediction models need appropriate internal, internal-external, and external validation. J Clin Epidemiol 2016;69:245–247. [published Online First: 2015/05/20].
- Takada T, Nijman S, Denaxas S, Snell KIE, Uijl A, Nguyen T-L, et al. Internal–external cross-validation helped to evaluate the generalizability of prediction models in large clustered datasets. J Clin Epidemiol 2021;**137**:83–91.
- Damen JAAG, Hooft L, Schuit E, Debray TPA, Collins GS, Tzoulaki I, et al. Prediction models for cardiovascular disease risk in the general population: systematic review. BMJ 2016;353:i2416.
- van Klaveren D, Balan TA, Steyerberg EW, Kent DM. Models with interactions overestimated heterogeneity of treatment effects and were prone to treatment mistargeting. *J Clin Epidemiol* 2019;**114**:72–83.
- van Klaveren D, Vergouwe Y, Farooq V, Serruys PW, Steyerberg EW. Estimates of absolute treatment benefit for individual patients required careful modeling of statistical interactions. J Clin Epidemiol 2015;68:1366–1374.
- Simon N, Friedman J, Hastie T, Tibshirani R. Regularization paths for cox's proportional hazards model via coordinate descent. J Stat Softw 2011;39:1–13.

- Tibshirani R, Bien J, Friedman J, Hastie T, Simon N, Taylor J, et al. Strong rules for discarding predictors in lasso-type problems. J R Stat Soc Ser B (Stat Methodol) 2012;74: 245–266.
- Vander-Weele TJ, Luedtke AR, van der Laan MJ, Kessler RC. Selecting optimal subgroups for treatment using many covariates. *Epidemiology* 2019;30:334–341.
- Heinze G, Schemper M. A solution to the problem of monotone likelihood in cox regression. *Biometrics* 2004;57:114–119.
- de Vries TI, Cooney MT, Selmer RM, Hageman SHJ, Pennells LA, Wood A, et al. SCORE2-OP risk prediction algorithms: estimating incident cardiovascular event risk in older persons in four geographical risk regions. Eur Heart J 2021;42:2455–2467.
- Trompet S, Postmus I, Slagboom PE, Heijmans BT, Smit RAJ, Maier AB, et al. Non-response to (statin) therapy: the importance of distinguishing non-responders from non-adherers in pharmacogenetic studies. Eur J Clin Pharmacol 2015;72:431–437.
- Micale Foody J, Rathore SS, Galusha D, Masoudi FA, Havranek EP, Radford MJ, et al. Hydroxymethylglutaryl-CoA reductase inhibitors in older persons with acute myocardial infarction: evidence for an aged€ statin interaction. J Am Geriatr Soc 2006;54: 421–430.
- Tilvis RS, Valvanne JN, Strandberg TE, Miettinen TA. Prognostic significance of serum cholesterol, lathosterol, and sitosterol in old age; a 17-year population study. Ann Med 2011;43:292–301.
- Armitage J, Baigent C, Barnes E, Betteridge DJ, Blackwell L, Blazing M, et al. Efficacy and safety of statin therapy in older people: a meta-analysis of individual participant data from 28 randomised controlled trials. *Lancet* 2019;**393**:407–415.
- Weverling-Rijnsburger AWE, Jonkers IJAM, van Exel E, Gussekloo J, Westendorp RGJ. High-Density vs low-density lipoprotein cholesterol as the risk factor for coronary artery disease and stroke in old age. Arch Intern Med 2003;163:1549.